Author's response to reviews

Title: microRNA-99a acts as a tumor suppressor and is down-regulated in bladder cancer

Authors:

Yongming Kang (yongming_kang@126.com)
Yougang Feng (uroygfeng@sina.com)
Yue He (yueyue0826@163.com)
Jun Liu (tj620080@163.com)
Bo Liang (liangbo_16@163.com)
Ping Yang (yo1104@sohu.com)
Zhou Yu (6988344@qq.com)

Version: 4 Date: 21 May 2014

Author's response to reviews: see over
Dear Dr. Partridge,

Thank you for your comments and suggestions on our manuscript titled “microRNA-99a acts as a tumor suppressor and is down-regulated in bladder cancer” (manuscript ID 1343897201117672). We have revised the manuscript according to the comments again and the revisions in the text are marked with red colour. The revisions and explanations are described as follows:

1. Authors' Contributions
We are sorry for the mistake including L.D. who did not contribute to the manuscript in the Authors' Contributions section. This is just a copy/paste mistake. We have deleted the description about L.D. in the revised manuscript.

2. Copy-editing
We have tried our best to correct the grammatical errors and the unclear sentences in the revised manuscript.

3. Competing Interests
We have changed this statement to “The authors declare that they have no competing interests” in the revised manuscript.

With the best regards,

Sincerely,

Yongming Kang
Department of Urology
Suining Central Hospital
127 Deshengxi Road, Chuanshan District
Suining 629000, China
Email: Kangyongming1979@163.com; yongming_kang@126.com
Tel: 86-0825-2292603